Author's response to reviews

Title: How Can Developing Countries Harness Biotechnology to Improve Health?

Authors:

   Abdallah S Daar (a.daar@utoronto.ca)
   Kathryn Berndtson (kathryn.berndtson@utoronto.ca)
   Deepa L Persad (deepa.persad@utoronto.ca)
   Peter A Singer (peter.singer@utoronto.ca)

Version: 5 Date: 27 November 2007

Author's response to reviews:

Dear Editors,

Please find attached our Final version of the manuscript: 'How can developing countries harness biotechnology to improve health?' with all of the requested formatting changes made.

Thank you very much for accepting our manuscript.

Best wishes,

The authors